Author's response to reviews

Title: Prognosis of critically ill cirrhotic versus non-cirrhotic patients: A comprehensive score-matched study

Authors:

Chung-Ming Fu (bugfu0624@gmail.com)
Chih-Hsiang Chang (franwisandsun@gmail.com)
Pei-Chun Fan (franwis1023@gmail.com)
Ming-Hung Tsai (tsccm@ms32.hinet.net)
Shu-Min Lin (smlin100@cgmh.org.tw)
Kuo-Chin Kao (kck0502@cgmh.org.tw)
Ya-Chung Tian (dryctian@cloud.cgmh.org.tw)
Cheng-Chieh Hung (cchung@cloud.cgmh.org.tw)
Ji-Tseng Fang (fangjits@cloud.cgmh.org.tw)
Chih-Wei Yang (cwyang@ms1.hinet.net)
Yung-Chang Chen (cyc2356@adm.cgmh.org.tw)

Version: 4
Date: 26 June 2014

Author's response to reviews: see over
Dear editors and reviewers,

We would like to submit our manuscript entitled “Prognosis of critically ill cirrhotic versus non-cirrhotic patients: A comprehensive score-matched study” by Chung-Ming Fu, Chih-Hsiang Chang, Pei-Chun Fan, Ming-Hung Tsai, Shu-Min Lin, Kuo-Chin Kao, Ya-Chung Tian, Cheng-Chieh Hung, Ji-Tseng Fang, Chih-Wei Yang, Yung-Chang Chen to BMC Anesthesiology for publication as research.

We appreciate very much the comments of the previous reviewers and the editorial office to improve this manuscript. We have amended the manuscript as suggested. The manuscript has been rewritten accordingly and the responses to the comments of the reviewers are described in a point-to-point manner. We hope that the revised manuscript will retain your attention and you will judge the revised manuscript to be suitable for publication in BMC Anesthesiology. We also declare that there is no conflict of interest, support or funding in this manuscript.

Thank you for reviewing our manuscript.

Yours sincerely,

Yung-Chang Chen

Department of Nephrology, Chang Gung Memorial Hospital.

199 Tung Hwa North Road, Taipei, 105 Taiwan

TEL: 886-3-3281200 ext 8181

FAX: 886-3-3282173.

E-mail: cyc2356@adm.cgmh.org.tw
Dear Editor

Thank you for the kind remind. All the requests were completed and listed as follows:

- Please remove your Short Title and Disclosure statement from your Main Title Page.
  ANS: The short title and disclosure were removed from the main title page.
- Please include the names of all ethical bodies that approved your study in the various centers involved. If you do not wish to list them all in the methods section, please include the list with the Acknowledgements section or as an additional file and refer to this in the methods section.
  ANS: The name of the ethical body was added in the “Methods” section.
- Please include a Conclusions section as the last section of the text. This should state clearly the main conclusions of the research and give a clear explanation of their importance and relevance. Summary illustrations may be included. Please add a Conclusions subheading to your concluding paragraph.
  ANS: Conclusions section was added as requested.
- Please include a Key Messages section after your Conclusions, these should be up to five bullet points summarising the main findings of your study.
  ANS: Key Messages section was added as requested.
- Please include a list of abbreviations used in the manuscript and their meanings, after the conclusion.
  ANS: The list of abbreviations was added as requested.
- Please include a competing interests section at the end of the manuscript, before the reference list. If the authors have no competing interests, please state: "The authors declare that they have no competing interests."
  ANS: Competing interests section was added as requested.
- Please include an Authors’ Contributions section at the end of the manuscript, before the reference list. We suggest the following kind of format (please use initials to refer to each author’s contribution): “AB carried out the molecular genetic studies, participated in the sequence alignment and drafted the manuscript. JY carried out the immunoassays. MT participated in the sequence alignment. ES participated in the design of the study and performed the statistical analysis. FG conceived of the study, and participated in its design and coordination and helped to draft the manuscript. All authors read and approved the final manuscript.”
  ANS: Authors contributions section was added as requested.
- If applicable, please include an acknowledgement section at the end of the manuscript before the reference list. Please acknowledge anyone who contributed towards the study by making substantial contributions to conception, design, acquisition of data, or analysis and interpretation of data, or who was involved in drafting the manuscript or revising it critically for important intellectual content, but who does not meet the criteria for authorship. Please also include the source(s) of funding for all authors. Authors should obtain permission to acknowledge from all those mentioned in the Acknowledgements.
  ANS: Acknowledgement section was added as requested.